Author’s response to reviews

Title: Lung function changes from childhood to adolescence: a seven-year follow-up study

Authors:

Pavilio Piccioni (ppiccioni@qubisoft.it)
Roberta Tassinari (roberta.tassinari@unito.it)
Aurelia Carosso (aur.carosso@gmail.com)
Carlo Carena (carlocarena@yahoo.it)
Massimiliano Bugiani (maxbugiani@libero.it)
Roberto Bono (roberto.bono@unito.it)

Version: 2 Date: 22 July 2014

Author’s response to reviews:

(1) line and page numbers were added

(2) Ethics Committee.

This study was approved by the Ethics Committee "St. Luigi Gonzaga Hospital" which was established according to the Legislative Decree n. 211 (24/6/2013). This Legislative Decree has attributed to the Ethics Committee "St. Luigi Gonzaga" the activities previously carried out by the Ethics Committee "ASL TO2" now dissolved.

The reference to the approval of the Ethics Committee is reported in the manuscript in the section: "Study Population".